Author's response to reviews

Title: Grey scale enhancement by a new self-made contrast agent in early cirrhotic stage of rabbit liver

Authors:

   Li Zhang (lilyzhang319_20@hotmail.com)
   Yun-You Duan (duanyy@fmmu.edu.cn)
   Ji-Kai Yin (yjkfmmuu@hotmail.com)
   Ji-Hong Cui (jihongcui@hotmail.com)
   Yong Zhang (zh_1015@163.com)
   Tie-Sheng Cao (caots@fmmu.edu.cn)

Version: 10 Date: 31 July 2007

Author's response to reviews: see over
ear editor,

We looked over the whole paper and made some corresponding changes about the format such as the affiliations about all authors. We removed the keywords section and added the figure legend and title and tables after the reference list. Thank you very much for your help for such a long time.

Thank you very much!

Yours sincerely,

Zhang Li